Author’s response to reviews

Title: Difference in vaginal microecology, local immunity and HPV infection among childbearing-age women with different degrees of cervical lesions in Inner Mongolia

Authors:

Jing-Jing Zheng (zhengjingjingcn@126.com)
Jing-Hui Song (jinghuisong@21cn.com)
Cong-Xiang Yu (yu_congxiang@sina.com)
Fei Wang (wangfei0903@21cn.com)
Peng-Cheng Wang (wang_pc7708@163.com)
Jing-Wei Meng (meng_jingwei@yeah.net)

Version: 3 Date: 25 Jul 2019

Author’s response to reviews:

Editor Comments:

Comments 1 Please revise your current Availability of Data and Materials statement to reflect one of the formats indicated in our submission guidelines: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

Reply: We have corrected 'Availability of data and material' section according to the submission guidelines.

Comments 2 Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests
Funding

Authors' contributions

Acknowledgements

Reply: We have corrected 'Declarations' section according to the guidelines.

Comments 3 Please can you add a "Declarations" heading to your Declarations section.

Reply: Yes. We have added the "Declarations" heading.

Comments 4 We note that Figure 1 is not currently cited in the text. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Reply: Thank you for your reminder. We have cited Figure 1 in the main text.

Comments 5 Currently the Abstract in your Manuscript differs from that which is submitted on our system. Please amend one of these so that they are the same.

Reply: Thank you for your reminder. We have corrected the Abstract.

Comments 6 Thank you for uploading your cover letter and financial disclosure. As these files are no longer required at this stage of the editorial process, please remove them from your submission.

Reply: Thank you for your reminder. We have removed these files.

Comments 7 Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. Please also ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Reply: Yes. We have revised the manuscript as you suggested. And we uploaded the revised manuscript as a single, final, clean version.

Reviewer reports:
Comments 1 In this correction, the authors have corrected about the point that I pointed out in the Tables.

Reply: Thank you for reviewing our study. And we appreciate your comments and help.